Be prepared to discuss the following topics in class tomorrow:

1. What commercial products use information you learned in COS 426.

How do they use it?

Do they make proper use of the information or could have done it better?

2. a) What was the most valuable thing you learned/did in 426.

b) How will you answer this question in 5 years?

3. Of the assignments in 426, which taught you the most

a) overall

b) per unit of your effort

4. Does it help to do graphics projects in teams?

5. Which of the lectures this semester will still be given

in the course 5 years from now?

Which of the lectures were not given 5 years ago?

6. How long will it be before there is an application program

that allows you to create a video like Toy Story

on your home computer?

What company will market it?

7. Overall, what changes would you make to the course?